Author's response to reviews

Title: Developing counseling skills through pre-recorded videos and role play: a pre- and post-intervention study in a Pakistani medical school.

Authors:

Noor Ahsen (drnahsen@gmail.com)
Syeda A Batul (atigabatul@yahoo.com)
Ahsen N Ahmed (drnahmed@hotmail.com)
Zakariya Imam (zakariya.imam@gmail.com)
Humaira Iqbal (humaira_nmc2468@hotmail.com)
Khayyam Shamshair (kshamshair@yahoo.com)
Hammad Ali (drhammadali@gmail.com)

Version: 2 Date: 11 June 2009

Author's response to reviews:

Upon reviewing my submission I realized that a reference had been incorrectly cited, which disturbed the sequence of the following references as well. I have revised this version with correction of reference numbers. The rest of the manuscript is unchanged.

Kindly replace the previously submitted version with the new one and review it.